Author’s response to reviews

Title: Role of P27-P55 operon from Mycobacterium tuberculosis in the resistance to toxic compounds

Authors:

María V Bianco (mbianco@cnia.inta.gov.ar)
Federico C Blanco (fblanco@cnia.inta.gov.ar)
Marina A Forrellad (mforrellad@cnia.inta.gov.ar)
Belen Imperiale (belen_imperiale@yahoo.com.ar)
Roxana V Rocha (vrocha@cnia.inta.gov.ar)
Laura I Klepp (lklepp@cnia.inta.gov.ar)
Nora Morcillo (nora_morcillo@yahoo.com.ar)
Angel A Cataldi (acataldi@cnia.inta.gov.ar)
Fabiana Bigi (fbigi@cnia.inta.gov.ar)

Version: 3 Date: 17 June 2011

Author’s response to reviews:

DDear Editor,

We would like to thank reviewer 2 for helping us with her carefully revision. The reviewer concerns are addressed point-by-point below and highlighted in the manuscript in yellow.

We hope that you will recommend publication of the revised manuscript in BMC Infection Diseases.

Sincerely,

Referee 2

Minor Essential Revisions

Pg 3, line 13: showen is showed.
It was corrected.

Pg. 4, line 4: Ref. no. 18 is not correct, but the correct reference is no. 10.
The reference number was changed as requested.

Pg. 7, line 2: tuberculosis is tuberculosis.
It was corrected.

Additional file 1: the value of H37Rv is not present.
The vale of H37Rv is 1 because data are relative to the expression in H37Rv.

Pg. 11, line 13: wirh is wrong.
It was corrected.
Pg. 14, line 5: show is showed.
It was corrected.
Pg. 21, Table 1: Reserpine 20 ng/ml is Reserpine 20 ug/ml (microgram/ml).
It was corrected.
Figures: Fig. 2 is in place of Fig. 3 and vice versa.
It was amended.